# Research Paper





# Investigating Psychic and Body Complications Caused by Inactivity Among Students of University During the COVID-19 Pandemic

Morteza Homayounnia Firouzjah<sup>†</sup>, Morteza Pourazar<sup>1</sup>

1. Department of Physical Education, Farhangian University, Tehran, Iran.



**Citation** Homayounnia Firouzjah, M., Pourazar, M. (2023). Investigating Psychic and Body Complications Caused by Inactivity Among Students of University During the COVID-19 Pandemic. *Journal of Practice in Clinical Psychology, 11*(2), 103-116. https://doi.org/10.32598/jpcp.11.2.703.3





**Article info:** 

Received: 10 Nov 2022 Accepted: 29 Nov 2022 Available Online: 01 Apr 2023

# **ABSTRACT**

**Objective:** The level of mobility and general health has decreased among students in virtual classes during the COVID-19 pandemic. The present study aims to investigate the mental and physical complications caused by inactivity among the students of Farhangian University during virtual classes.

**Methods:** This was survey research with a cross-sectional design. The statistical population included all male and female students studying at Farhangian University of Mazandaran Province, Iran. According to the determination of the sample size based on the Morgan table, 475 students, consisting of 214 females and 261 males, were randomly selected as the statistical sample of the study. The research instruments include the international physical activity questionnaire, Saehan Caliper (SH5020), the Coopersmith self-esteem scale (1967), the Beck depression questionnaire, and the Nordic skeletal and muscular disorders questionnaire. To analyze the data, we used the independent sample t test to compare the two groups. All analyses were conducted using the SPSS software, version 24.

Results: The research findings showed that the average weekly activity level among women was 634±281 MET/min, and the average weekly activity level among men was 472±231 MET/min. In terms of fat percentage by gender, men's average fat percentage was 21.47%±4.74%, and women's average fat percentage was 31.55%±4.37%. Meanwhile, the self-esteem scores of male and female students were obtained at 29.72 and 29.43, respectively. The difference between the two was considered significant as P<0.05. In addition to these findings, the correlation between self-esteem and activity level was +0.41. On the other hand, 25(67%) female students and 12(32%) male students suffered from high depression.

Conclusion: Concerning students' skeletal-muscular disorders, findings proved that both genders suffered from physical complications during virtual classes. This study suggests increasing the level of physical activity to reduce body fat mass, increase mental health, and reduce skeletal disorders, which can be properly accomplished through university planning and prioritizing the health of male and female students.

#### **Keywords:**

Activity, Mental and physical complications, Farhangian University, COVID-19

#### \* Corresponding Author:

Morteza Homayounnia Firouzjah, Assistant Professor.

Address: Department of Physical Education, Farhangian University, Tehran, Iran.

**Tel:** +98 (910) 6803022

E-mail: Mortezahomayoun@cfu.ac.ir

# **Highlights**

- The self-confidence level of male students was higher than that of female students during COVID-19.2.
- The fat percentage of female students was higher than that of male students during COVID-19.3.
- Regarding the amount of skeletal-muscular disorders of the students, the findings showed that the highest frequency of pain, discomfort and numbness in the upper limbs of all students was more than the lower limbs

# **Plain Language Summary**

Determining the prevalence and pattern of mental disorders and musculoskeletal pain is the first step in the prevention, diagnosis, and treatment of such disorders, even though a targeted and reliable study in this field has not been conducted during the COVID-19 pandemic when students had to use the virtual space for classrooms instead of physically attending the class. Therefore, through this research results, appropriate solutions and detailed plans can be devised to alleviate the mental and physical complications of students.

#### 1. Introduction

ndustrial development, the upsetting situation of the COVID-19 pandemic, virtual classes, and life mechanization have majorly affected individuals' lifestyles and have brought sedentary lifestyles to societies (Nayak et al., 2022). This inactivity in students, as the active class of future society, can cause structural and strategic problems for any country (Yang et al., 2022). Therefore, inactivity should be considered a serious problem for the future of a country (Montero-Simó et al, 2022). The role of comprehensive education is to provide education for all humanistic dimensions, which includes the intellectual aspect of the individual along with all psychological and physical aspects (Yuan et al., 2022). This type of comprehensive and inclusive education is beneficial for educational environments, such as universities and colleges (Demchenko et al., 2021). Today, to better adapt to the surrounding environment, individuals in society require a balance of physical readiness and body composition (Mazza et al., 2020), and if they do not have favorable conditions in terms of physical condition and body composition, they usually become aloof, pessimistic, and isolated. In other words, these individuals will not have proper mental balance (Izquierdo, 2005). Obesity is a chronic condition that occurs as a result of intervention in individuals' genetics or their environment, which is affected by society, culture, psychological, metabolic, biochemical, and genetic conditions (Kim et al., 2021). Findings have demonstrated that male students living in urban areas are more likely to be obese (39.4%) compared to male students who live in the suburbs (35.5%). Likewise, 20.6%

of urban female students were exposed to obesity compared to suburban females (19.1%). The increasing prevalence of obesity and overweight among children and adults in the United States of America is a warning for doctors and public health officials (Halpern et al., 2021). In different countries, obesity is probably caused by a decrease in physical activity and an inappropriate lifestyle. High levels of health and hygienic indices and physical capabilities reflect the health and potentialities of a society (Wang et al., 2021). It has been well proven that the level of physical activity begins to decrease during adulthood and with age which is associated with weight gain. The evidence that proves the relationship between physical activity in childhood and adolescence and inactivity in individuals in the form of longitudinal studies is rare or does not exist at all. A longitudinal study of 5700 men and women found a link between childhood activity and obesity in adulthood. Those individuals who do not engage in sports activities and then become overweight may be genetically predisposed to this factor because both physical activity and body size are affected by genetic factors (Chaput et al., 2020). In the last few years, inactivity has become widespread in a way that since the 1990s, it has been proposed as one of the main factors of death because of cardiovascular diseases (Ralapanawa & Sivakanesan, 2021). Those individuals who have done intense physical activity enjoy better health conditions compared to individuals with moderate and light activities (Rosenberger et al., 2019). Studies on the level of inactivity and public health have shown that the increase in inactivity is associated with an increase in obesity, as well as a decrease in physical activity with a decrease in general health (McCoy & Morgan, 2020). Research related

to the level of mobility in a different stratum of society has progressed such that the effect of the education level on food intake, obesity, and other health risk factors (mobility level) has changed over time (Su et al., 2022). Additionally, regular physical activity leads to an increase in physical readiness, and a significant relationship exists between high physical readiness and high self-esteem (Reddy et al., 2021). Some researchers have shown that yoga practice and physical exercises strongly influence individuals' personalities, coping skills, and cognitive performance (Farhang et al., 2022; Pascoe et al., 2021; Sinha & Kumari, 2022). Compared to other groups, Yoga practitioners had higher "Sattva Guna" (balance feature) and preferentially, employed brain regions associated with self-regulation and inhibitory control. Also, other researchers have stated that physical activity is essential for children's current and future health, though most of them do not do 60 min of moderate to intense daily physical activity (Ito et al., 2021; Tandon et al., 2021).

Programs related to physical activity are among the most common ways to increase self-esteem (Mazereel et al., 2021). In addition, compared to complex and heavy activities, simple activities, such as aerobic sports, have the greatest effect on self-esteem (Ryan, 2008). It has been proved that low self-esteem is related to depression, low mental health, and less progress in education (Sánchez-SanSegundo et al., 2022). Self-esteem is related to dissatisfaction with body image in obese individuals who follow weight loss diets (Chang & Kim, 2022). It has been well proven that a decrease in individuals' self-esteem is related to a decrease in their general health level (Hajek & König, 2019). The relationship between self-esteem and obesity has not been well-proven yet (Fields et al., 2021). Although challenges related to selfesteem have significant results on an individual's health, because of incomplete results, it is difficult to argue that low self-esteem is a consequence rather than a cause (Bleidorn & Schwaba, 2018). The decrease in muscle volume and general excessive thinness caused by lack of movement endangers the health of the body's skeleton (Jestratijevic et al., 2022). As a result of excessive muscle wasting, the trunk cannot perform its functions in maintaining the body and preserving its natural alignment, thereby resulting in bad standing, bad sitting situations, and overall wrong movement habits (Wijngaarde et al., 2020). This makes the spine and chest unable to grow normally and remain in a normal state. In addition to the ugliness and deformity of the body, the unnatural curvature of these organs causes the blood flow and breathing to be performed improperly and unnaturally (Hast & Garrison, 2000).

The most important consequences of a lack of movement are illness and reduction of muscle volume and strength (Bonnet et al., 2019). The above items are related to each other and a direct relationship exists between the two factors of muscle cross-section and the amount of power that a muscle can generate. On the other hand, maintaining the skeletal balance of the body is the responsibility of the muscles, especially the amount of strength and power of each muscle. Those individuals who do not have enough physical preparation (readiness), get tired sooner during physical activities. Muscle fatigue in the body will naturally reduce the physical ability and the power level that the muscle can represent during working situations (Blocquiaux et al., 2020). In other words, working with tired muscles is the same as working with weak muscles, and the adverse effects that occur as a result of muscle weakness in a person are also similar to those of working in extreme fatigue conditions. Fatigue and posture may have a cause-andeffect relationship. In this way, the presence of fatigue caused by other factors, such as physical activities in the body, can be influential in disrupting the balance of an appropriate posture. On the other hand, the lack of a proper body posture is a reason for causing more fatigue; the more the body size is out of proportion and balance, the more energy is required to keep it straight because the muscles related to the way the body is positioned to maintain its balance have less mechanical merit. On the other hand, these muscles should be involved in activities that will cause body fatigue (Gaudiino & Di Stefano, 2021). If the child's daily activity is less than normal, the weight of other parts of their body will gradually decrease and the weight and volume of the subcutaneous fat tissue will increase, eventually leading to the child's obesity. Because subcutaneous fat does not equally accumulate in all parts, and in some parts, such as around the abdomen, hips, and the middle part of the body is more than the organs (arms and legs), the child's body position becomes abnormal (Lemaitre et al., 2021).

According to the reports of the Centers for Disease Control and Prevention in 2018, a total of 29% of school students did not pay attention to physical education classes (Brustio et al., 2018). Meanwhile, another study stated that students with daily physical activity demonstrate higher academic performance (Páez-Maldonado et al., 2020). Physical activity is a behavior that has many proven health benefits, along with being one of the most effective ways to prevent chronic diseases, such as coronary heart disease and diabetes (Speelman et al., 2011).

This study aims to determine the frequency and complications (mental and physical) caused by inactivity in students of Farhangian University during virtual classes. The main objective is to measure the level of physical activity and the effects of inactivity on mental and physical factors among students studying at Farhangian University of Mazandaran Province, Iran.

#### 2. Materials and Methods

This was a survey study with a cross-sectional design. The statistical population includes all male and female students studying at Farhangian University of Mazandaran Province, Iran. The students were selected randomly and clustered in a way that initially the campuses of Mazandaran Province were divided based on the city and their population, and then students of the cities were randomly selected. The number of samples was selected based on the population; therefore, more samples were allocated to campuses with larger populations. In the following, based on the statistics received from Farhangian University of Mazandaran Province, Iran, according to the sample size based on the Morgan table, a total of 475 students consisting of 214 females and 261 males were randomly selected as the statistical sample of the research.

#### Research instruments

#### Amount of physical activity

After obtaining the informed consent letter, the amount of physical activity was calculated using the international physical activity questionnaire (IPAQ) (Craig et al., 2003). In this questionnaire, the physical activities performed by the subjects during the last week were asked, and activities performed for more than 10 min were recorded. These activities included job activities, moving manner, doing household chores, and leisure activities. This questionnaire asks about the amount of intense and moderate physical activity and walking during the last week. According to the scoring protocol of the IPAQ questionnaire, the amount of physical activity of a person can be extracted and reported in two ways as follows:

The total amount of physical activity of the individual during the last week in terms of met-min/week

MET (Tan et al., 2021) is a unit used to estimate energy consumption as a result of physical activity. The value of one MET is approximately equal to the amount of energy consumed by a person at rest. All physical activities can be classified as multiples of the amount of energy consumption in a resting state. In this questionnaire, 3.3

METs are considered for walking, 4 METs for moderate physical activity, and 8 METs for intense physical activity. To calculate the total amount of physical activity in a week, the amount of walking (MET×min×day) should be added together with the amount of moderate physical activity (MET×min×day) and the amount of intense physical activity (MET×min×day) in the last week.

Classification of individuals' physical activity in three levels: Low, medium, and high

High physical activity means an individual has intense physical activity at least 3 days a week and a total of at least 1500 MET-min, or they do any combination of intense, moderate, walking activities for 7 or more days, with a total of at least 3000 MET-min per week. Moderate physical activity means an individual has at least 20 min of intense physical activity 3 days a week or more, or 5 days or more per week, the individual has at least 30 min of intense or moderate activity or walking. Low physical activity means an individual does not report any activity or the reported physical activities do not meet the criteria of high or moderate physical activity (Fogelholm et al., 2006). In the present study, after conducting preliminary studies on the necessity of researching 2 categories of low and moderate activity intensity, the high physical activity category of the intended samples was disregarded; therefore, the category of high activity level was excluded from the study, and the current study was based on the level of activity and the percentage of body fat, as well as the level of self-esteem of two classes with a low and moderate level of physical activity.

To determine the validity of the questionnaire, we used the content validity method. The reliability was measured by the test-retest method, and the correlation coefficient was obtained at 0.62 for the awareness and attitude section and 0.74 for the performance section (Moghaddam et al., 2012).

#### Fat percentage

The participants' fat percentage was measured with a Saehan (SH5020) fat meter (caliper) made in England at 3 points of the body (men: Chest, thighs, and abdomen; women: Triceps, upper arms, and thighs) (Onsori & Galedari, 2015). To increase the reliability of the subcutaneous fat measurement process, each body part was measured 3 times with a specific time interval and all measurements were performed on the right side of the body (Soltani et al., 2018). To determine the fat percentage of the subjects, the measured values were put into Jackson Pollock's fat measurement formula and the fat

percentage was calculated. The standing height of the participants was measured using the height meter model 216 (SECA). For this purpose, the subjects stood in a manner that the rear part of their shoulders touched the height-measuring device. They kept their hands next to their body and close to their feet. The weight of each person was measured using a SECA model scale.

# Coopersmith self-esteem questionnaire-short form

The Coppersmith self-esteem scale (1967) (Morrison et al., 1973) is a 58-item self-report, pencil-paper questionnaire, and 8 items are lie-detectors while the other 50 items are divided into 4 subscales, namely general self-esteem, social self-esteem, family self-esteem, and educational self-esteem. The purpose of this questionnaire is to evaluate students' self-esteem. This test has different forms. The original test was primarily designed for 8–15-year-old subjects (form A, or school form); however, a later revision was designed for subjects over 16 years of age (form C, or adult form). Some items were rewritten to adapt the original form for adults' use (form C; for instance, children were replaced with individuals, and schools with work). In addition, a short form of the test (form B, Coopersmith, 1987) exists which includes 25 items and is extracted from the 50-item scale. Coopersmith designed this form as an alternative for limited times. The reliability coefficient of this test is reported at 0.77. The Coopersmith self-esteem scale is standardized in Iran (Narimani & Mousazadeh, 2010).

#### Depression scale

The Beck depression questionnaire was first developed by Beck et al. in 1961 (Jackson-Koku, 2016). In 1996, Beck et al. made a major revision to cover a wider range of symptoms and provided more consistency with the diagnostic criteria for depressive disorders in DSM-IV. The Beck depression questionnaire is a type of selfassessment test and can be completed in 5 to 10 min. The test consists of a total of 21 items related to different symptoms, in which the participants must answer on a 4-point scale from 0 to 3. These articles cover various areas, such as sadness, pessimism, feelings of incapability and failure, guilt, sleep disturbances, loss of appetite, self-loathing, and so on. Accordingly, 2 items are devoted to emotion, 11 items to cognition, 2 items to overt behaviors, 5 items to physical signs, and 1 item to interpersonal semiotics. Thus, this scale determines different degrees of depression from mild to very severe, and its scores range from a minimum of 0 to a maximum of 63. The Cronbach  $\alpha$  coefficient of this questionnaire was reported at 0.84 (Farshchi et al., 2018).

#### Musculoskeletal disorders

To examine skeletal-muscular ailments by the doctor, the Nordic questionnaire was employed. This tool is a standardized questionnaire for examining disorders and diseases caused by working and daily affairs (López-Aragón et al., 2017). The reliability of this scale has been reported at 0.73 using the Cronbach  $\alpha$  method (Namnik et al., 2016).

#### Data analysis

To analyze the data, the independent sample t-test statistical method was applied to compare the two groups. All analyses were carried out using the SPSS software, version 24.

## 3. Results

According to the research findings (Table 1) and the ratio of male and female students at Farhangian University of Mazandaran Province, Iran, 59% of the participants were male and 41% were female. The findings of this study showed that 348(73%) students had low physical activity and 127(27%) had sufficient physical activity.

Table 1. Demographic characteristics of participants

Attributes —	Mean±SD					
	Male Students	Female Students	Total			
Age (y)	20.47±4.32	20.67±3.94	22.27±3.64			
Height (cm)	172.19±4.25	163.57±4.64	169±3.98			
Weight (kg)	70.14±8.25	59.91±8.32	66.86±7.61			
Body mass index (kg/m²)	21.64±1.26	22.13±1.38	21.76±1.44			
			PRACTICE in			

CLINICAL PSYCHTLOGY

Table 2. Comparison of body fat percentage of male and female students based on their activity level

Gender	a astrota.	Fat Percentage		
	Activity	Mean±SD	No.	- t
Male	Active	20.19±4.62	71	
	Inactive	21.54 <b>±</b> 4.43	190	2.14
	Total	21.47 <b>±</b> 4.74	261	
	Active	30.15±4.43	52	
Female	Inactive	31.64±4.18	162	2.52
	Total	31.55 <b>±</b> 4.37	214	

CLINICAL PSYCHTLOGY

Also, among 348(73%) sedentary students, 138(40%) were women while 210 (60%) were men. On the other hand, among 127(27%) students with sufficient mobility, 109(85%) were men and 18(15%) were women.

Also, the research findings from examination regarding the pervasiveness of obesity in students showed that the average fat percentage of the subjects was 25.19%±7.44%, while in terms of gender, the average fat percentage of men was obtained at 21.47±4.74% and the average fat percentage of women was reported at 31.55±4.37%. This difference between men and women is considered significant with P<0.05. Also, the average level of physical activity of all students was 578±284 Met/min. The present study showed that the average weekly activity level among women was 634 Met/min with a standard deviation of ±281 while the average weekly activity level among men was 472 Met/min with a standard deviation of ±231. In addition to these find-

ings, the correlation between fat mass percentage and activity level was -0.78 (Table 2).

Also, regarding students' level of self-esteem, our findings confirmed that 356(74%) of all male and female students had high self-esteem, and 119(26%) of them had low self-esteem. Among this 74%, the share of female students was 164(46%) and the share of male participants was 192 (54%). On the other hand, among those who had low self-esteem, female students accounted for 52(42%) and male participants accounted for 69 (58%). However, no significant difference was found between the level of activity and self-esteem. Information regarding students' level of activity and their self-esteem is provided in Table 3.

Accordingly, the total self-esteem of the participants was 28.6±5.7. The average score of the students with high self-esteem was 30 and the average score of the students with low self-esteem was 21. Also, the self-es-

Table 3. Comparison of self-esteem of male and female students based on their activity level

Gender	A akin ika	Self-esteem		
	Activity —	Mean±SD	No.	- t
Male	Active	30.82±5.47	192	
	Inactive	29.45±5.16	69	1.78
	Total	29.72±4.37	261	
	Active	29.46±6.46	164	
Female	Inactive	28.48±6.37	50	1.27
	Total	29.43±6.52	214	

PRACTICE IN CLINICAL PSYCH OLOGY

Table 4. Comparison of the level of depression of male and female students based on their activity level

Gender	a attack.	Depression		
	Activity	Mean±SD	No.	- t
Male	Active	20.64±6.31	192	
	Inactive	27.39±6.29	69	2.48
	Total	25.71±6.43	261	
Female	Active	20.52 <b>±</b> 5.67	164	
	Inactive	29.61±6.44	50	2.04
	Total	25.65±6.30	214	

PRACTICE IN CLINICAL PSYCH®LOGY

teem scores of male and female students were obtained at 29.72 and 29.43, respectively. The difference between the two was considered significant as P<0.05. In addition, the findings suggested that the correlation between self-esteem and activity level was +0.41.

Also, regarding the level of depression among students, the findings demonstrated that 438(82%) of all male and female students had low depression and 37(18%) of them had high depression. Among this 82%, the share of female students was 203(45%) participants and the share of male individuals was 236(55%). On the other hand, female students accounted for 25(67%) and male subjects accounted for 12(32%) of high depression. However, no significant difference was found between the level of activity and depression. Table 4 provides information about the level of activity and self-esteem of students.

According to Table 5, regarding musculoskeletal disorders of the students, the findings indicate that the highest frequency of pain, discomfort, and numbness in the last 12 months in active male students were related to wrists and hands (n=17), in the last 7 days related to back (n=16), and the most skeletal pain that made them stop physical activity in the last 12 months was a pain in the knees (n=21). In addition, in sedentary male students, the highest frequency of pain, discomfort, and numbness in the last 12 months was related to the back (n=42), during the last 7 days related to the thigh (n=64), and the most skeletal pain made them stop physical activity in the last 12 months was due to back pain (n=74).

In contrast, in female students, the highest frequency of pain, discomfort, and numbness in the past 12 months in active female students were related to the neck (n=15), in the last 7 days related to the wrists and hands (n=22), and the most skeletal pain that in the last 12 months made

them quit physical activity was related to shoulder pain (n=31). Also, in sedentary female students, the highest prevalence of pain, discomfort, and numbness in the last 12 months was related to the back (n=72), during the last 7 days related to the shoulders (n=103), and the most skeletal pain in the last 12 months that caused them to quit physical activity was related to back pain (n=108).

## 4. Discussion

The present study aimed to investigate the mental and physical complications caused by inactivity among the students of Farhangian University during virtual classes as a result of the COVID-19 pandemic. Determining the prevalence and pattern of mental disorders and musculo-skeletal pain is the first step in the prevention, diagnosis, and treatment of such disorders, even though a targeted and reliable study in this field has not been conducted during the COVID-19 pandemic when students had to use the virtual space for classrooms instead of physically attending the class. Therefore, through this research results, appropriate solutions and detailed plans can be devised to alleviate the mental and physical complications of students.

The results of this study suggest that the prevalence of inactivity among students is high, and about 73% of all individuals did not participate in any of the intense and moderate activities, while in other countries, such as Saudi Arabia, the prevalence of physical inactivity comprises more than 43% of society (Al-Hazzaa, 2007). In the United States, the prevalence of overweight is 36% and obesity is 21% (Davis & Gergen, 1994; Gordon-Larsen, 2001). This amount is reported at 18% (Ramos de Marins, 2001) in Ireland and 33% (McCarthy et al, 2002) in Brazil. Also, the present findings showed that female students had less physical activity (73%) compared to male students (51%).

Table 5. Comparing the degree of musculoskeletal disorders of male and female students based on their activity level

Gender Activity				Number of People With Skeletal and Muscular Disorders					_		
	Activity	Type of Disorder	Neck	Shoulders	Elbow	Wrists and hands	Back	Waist	leg	Knees	Feet
	Active	Pain, discomfort, and numbness during the past 12 months	2	5	2	17	15	11	2	8	9
		Pain, discomfort, and numbness during the past 7 days	5	14	15	14	16	14	10	2	3
Male		Absence of daily activities during the past 12 months	5	14	11	11	9	5	17	21	14
students		Pain, discomfort, and numbness during the past 12 months	24	17	25	32	16	42	17	24	34
	Inactive	Pain, discomfort, and numbness during the past 7 days	21	30	12	51	41	38	64	51	63
		Absence of daily activities during the past 12 months	52	42	47	32	74	42	24	31	20
	Active	Pain, discomfort, and numbness during the past 12 months	15	10	12	6	8	10	14	12	9
		Pain, discomfort, and numbness during the past 7 days	3	14	15	22	12	17	11	13	10
Female students Inactive		Absence of daily activities during the past 12 months	21	31	12	18	20	14	17	18	22
		Pain, discomfort, and numbness during the past 12 months	61	25	47	29	72	62	71	53	37
	Inactive	Pain, discomfort, and numbness during the past 7 days	37	103	62	89	16	42	37	89	62
	Absence of daily activities during the past 12 months	81	62	37	54	64	108	82	39	49	

PRACTICE IN CLINICAL PSYCH®LOGY

Male students demonstrated more physical activity at all levels compared to female participants. Regarding the number of female students to male students, the current research results indicated that the activity level of female students was much lower than male students, which could be because of the COVID-19 period, social restrictions, and more closure of sports clubs for girls. On the other hand, the lack of physical activity facilities for both male and female students had a significant impact on their lack of exercise.

Furthermore, researchers have stated that highly-educated individuals have a low level of activity, while other researchers (Lewis, 2005) have shown that the amount of physical activity decreases in those with low education. Moreover, based on a study (Hajian-Tilaki & Heidari, 2007), no significant relationship was found between physical activities and education levels, which is consis-

tent with the study of other researchers (Trockel, Barnes, & Egget, 2000; Wilsgaard et al., 2005). Meanwhile, the average level of physical activity in this study showed a significant difference between the level of physical activity of men and women, i.e. men who participated in this study had a higher level of physical activity compared to women. Moreover, during the last 3 decades, a significant increase in obesity among children and adults has been observed (Carter et al., 2011). This issue has spread to the point that with the increase in individuals' education level, their body fat percentage increases as well (Marmot, 2003). Simultaneously, other researchers have reported results contrary to this finding (Morrill et al., 1991). In the present study, educated subjects did not have a very high amount of fat mass. The prevalence of obesity in Venezuela is 74% for men and 56% for women (Campos et al., 2003), which is consistent with the results of this study regarding the difference between women's and men's levels of physical activity. However, this rate in Palestine is 48% for men and 65% for women (Campos et al., 2003), which shows contradictory results to our findings. According to all the findings, the present study showed that the difference in fat mass between sedentary and sufficiently active participants was significant. This means that sedentary subjects had a higher fat mass. Another research has shown that the low participation of individuals in educational programs is related to a decrease in their self-esteem, demonstrating a positive correlation (Suss et al., 1996).

Accordingly, by reducing study hours, students' self-esteem decreases. This is in line with the results of the present study because the students of Farhangian University had high self-esteem. Therefore, it can be concluded that an individual's level of self-esteem probably increases with the increase in their education level. Moreover, no significant relationship was observed between the level of activity and high/low self-esteem which means that both low-activity and sufficiently active groups showed high self-esteem scores. In this regard, other research findings found a significant relationship between self-esteem and obesity (French et al., 1995), which is consistent with the findings of our study. In contrast to these results, another study proved that obese women had lower self-esteem (Pesa, Syre, & Jones, 2000). This could be because by the increase in weight, the amount of mobility reduces and the individual will have fewer social connections and less participation in daily activities, thereby possibly reducing their self-esteem. According to the present study, given the high educational level among the participants and their status at high social and cultural levels, low mobility could not impact their self-esteem. Based on the results (Scherrer & Preckel, 2019), self-esteem does not change significantly with changes in the amount of fat mass because no statistically significant difference exists between sedentary and physically active students in their body fat mass. However, in the opposite claim (Anderson et al., 2006; Davis & Gergen, 1994; Guinn, Semper, Jorgensen, & Skaggs, 1997; Wang et al., 2009), researchers have shown an inverse relationship between individuals' body weight and self-esteem; that is, with a decrease in body weight, the amount of self-esteem increases, and with an increase in body weight, self-esteem decreases. On the other hand, other scientists (Childress et al., 1993) claimed that the level of self-esteem among overweight children was significantly lower than their normal counterparts, which is not consistent with the findings of this study as a significant difference was not observed between the amount of fat mass as well as the activity level with the level of self-esteem. The reason for this difference could be attributed to the age differences between children and adults.

Concerning the level of depression of Farhangian University students, the findings showed that the level of depression of female students during the COVID-19 pandemic and virtual classes was higher compared to male subjects, but this was relatively small and it can be expected that with the increase of psychological and counseling interventions, the case would be reduced. On the other hand, no significant difference was found between the level of activity and depression.

Also, regarding the degree of the musculoskeletal disorders of the students, findings suggested that among active male students, the highest frequency of pain, discomfort, and numbness in the last 12 months were related to the wrists and hands (n=17), in the last 7 days related to back (n=16), and the most skeletal pain that caused them to stop their physical activity in the last 12 months was the pain in their knees (n=21). Also, in sedentary male students, the highest prevalence of pain, discomfort, and numbness in the last 12 months was related to the back (n=42) and in the last 7 days, it was related to the thigh (n=64); meanwhile, the most skeletal pain in the last 12 months which made them quit physical activity was related to back pain (n=74). On the other hand, in female students, the highest prevalence of pain, discomfort, and numbness in the past 12 months was related to the neck (n=15) and during the last 7 days, it was related to wrists and hands (n=22); additionally, the most skeletal pain that caused them to leave physical activity in the last 12 months was related to shoulder pain (n=31). Also, in sedentary female students, the highest frequency of pain, discomfort, and numbness in the last 12 months was related to the back (n=72), and in the last 7 days, it was related to the shoulders (n=103); also, the most skeletal pain in the last 12 months which made them quit physical activity was related to back pain (n=108).

The findings of this study are similar to the global statistics of the COVID-19 pandemic and its impact on individuals' physical and mental factors. The complications caused by the COVID-19 pandemic, as a result of the presence of the students in person, can have negative effects on students. Therefore, it can be concluded that even students and their level of understanding of the benefits of physical activity cannot prevent their mental and physical problems. It is expected that students who use modern scientific resources and are relatively aware of the dangers of obesity and inactivity are not exposed to such injuries; however, in practice, during virtual classes, students of Farhangian University were not sufficiently active. As a result, physical and mental complications were observed among this population. Also, the results related to the degree of psychological factors (self-esteem and depression) showed that in terms of mental health, both female and male students were not in satisfactory condition due to a lack of training and other related factors. Hence, during the COVID-19 pandemic, given the health restrictions and government policies regarding quarantines and the pandemic, and on the other hand, virtual classes and using communication devices for attending class, students did not have enough time for exercise. Students did not enjoy physical activity because physical activity was not a priority for students of Farhangian University during the COVID-19 pandemic.

This study suggests increasing the level of physical activity to reduce body fat mass, enhance mental health, and reduce skeletal disorders, which can be properly accomplished through organized university planning and prioritizing the health of male and female students.

#### **Ethical Considerations**

# Compliance with ethical guidelines

All methods were carried out following relevant guidelines and regulations, also all experimental protocols were approved by the Ethics Committee of the non-interventional, non-pharmaceutical clinical research of Farhangian University on April eleventh, 2022. The study procedures were explained to the participants and informed consent was obtained from all subjects and their parents before starting the study.

#### **Funding**

This research did not receive any specific grant from funding agencies in the public, commercial, or not-for-profit sectors.

#### Authors' contributions

Conceptualization, data curation, investigation, methodology, resources, supervision, validation, visualization, and writing (reviewing and editing): The both authors; Project administration, resources and validation: Morteza Homayounnia Firouzjah.

#### Conflict of interest

The authors declare no conflict of interest.

#### Acknowledgments

The authors would like to thank all students of Farhangian University of Mazandaran Province, Iran, who kindly cooperated in the present study. We also thank and appreciate the support of the Farhangian University of Mazandaran Province, Iran, in carrying out this project.

#### References

- Al-Hazzaa, H. M. (2007). Health-enhancing physical activity among Saudi adults using the International Physical Activity Questionnaire (IPAQ). Public Health Nutrition, 10(1), 59-64. [DOI:10.1017/S1368980007184299] [PMID]
- Anderson, S. E., Cohen, P., Naumova, E. N., & Must, A. (2006). Association of depression and anxiety disorders with weight change in a prospective community-based study of children followed up into adulthood. *Archives of Pediatrics & Adolescent Medicine*, 160(3), 285-291. [DOI:10.1001/archpedi.160.3.285] [PMID]
- Bleidorn, W., & Schwaba, T. (2018). Retirement is associated with a change in self-esteem. *Psychology and Aging*, 33(4), 586–594. [DOI:10.1037/pag0000253] [PMID]
- Blocquiaux, S., Gorski, T., Van Roie, E., Ramaekers, M., Van Thienen, R., & Nielens, H., et al. (2020). The effect of resistance training, detraining and retraining on muscle strength and power, myofibre size, satellite cells and myonuclei in older men. *Experimental Gerontology*, 133, 110860. [DOI:10.1016/j.exger.2020.110860] [PMID]
- Bonnet, N., Bourgoin, L., Biver, E., Douni, E., & Ferrari, S. (2019). RANKL inhibition improves muscle strength and insulin sensitivity and restores bone mass. *The Journal of Clinical Investigation*, 129(8), 3214-3223. [DOI:10.1172/JCI125915] [PMID] [PMCID]
- Brustio, P. R., Moisè, P., Marasso, D., Alossa, D., Miglio, F., & Mulasso, A., et al. (2018). Participation in a school-based walking intervention changes the motivation to undertake physical activity in middle-school students. *PloS One*, 13(9), e0204098. [DOI:10.1371/journal.pone.0204098] [PMID] [PMCID]
- Campos, G., Ryder, E., Diez-Ewald, M., Rivero, F., Fernández, V., & Raleigh, X., et al. (2003). [Prevalence of obesity and hyperinsulinemia: Its association with serum lipid and lipoprotein concentrations in healthy individuals from Maracaibo, Venezuela (Spanish)]. *Investigacion Clinica*, 44(1), 5-19. [PMID]
- Carter, P. J., Taylor, B. J., Williams, S. M., & Taylor, R. W. (2011). Longitudinal analysis of sleep in relation to BMI and body fat in children: The FLAME study. BMJ (Clinical research ed.), 342, d2712. [DOI:10.1136/bmj.d2712] [PMID] [PMCID]
- Chang, A. K., & Kim, S. H. (2022). Predictors of weight-control behavior in healthy weight and overweight Korean middleaged women. *International Journal of Environmental Research* and Public Health, 19(12), 7546. [DOI:10.3390/ijerph19127546] [PMID] [PMCID]
- Chaput, J. P., Willumsen, J., Bull, F., Chou, R., Ekelund, U., & Firth, J., et al. (2020). 2020 WHO guidelines on physical activity and sedentary behaviour for children and adolescents aged 5-17 years: Summary of the evidence. *The International Journal of Behavioral Nutrition and Physical Activity*, 17(1), 141. [DOI:10.1186/s12966-020-01037-z] [PMID] [PMCID]
- Childress, A. C., Brewerton, T. D., Hodges, E. L., & Jarrell, M. P. (1993). The Kids' Eating Disorders Survey (KEDS): A study of middle school students. *Journal of the American Academy of Child & Adolescent Psychiatry*, 32(4), 843-850. [DOI:10.1097/00004583-199307000-00021] [PMID]

- Craig, C. L., Marshall, A. L., Sjöström, M., Bauman, A. E., Booth, M. L., & Ainsworth, B. E., et al. (2003). International physical activity questionnaire: 12-country reliability and validity. *Medicine and Science in Sports and Exercise*, 35(8), 1381-1395. [DOI:10.1249/01.MSS.0000078924.61453.FB] [PMID]
- Davis, H., & Gergen, P. J. (1994). Self-described weight status of Mexican-American adolescents. The Journal of Adolescent Health: Official Publication of The Society for Adolescent Medicine, 15(5), 407-409. [DOI:10.1016/1054-139X(94)90265-8] [PMID]
- Demchenko, I., Maksymchuk, B., Bilan, V., Maksymchuk, I., & Kalynovska, I. (2021). Training future physical education teachers for professional activities under the conditions of inclusive education. BRAIN. Broad Research in Artificial Intelligence and Neuroscience, 12(3), 191-213. [DOI:10.18662/brain/12.3/227]
- Farhang, M., Miranda-Castillo, C., Behrens, M. I., Castillo, E., Mosquera Amar, S., & Rojas, G. (2022). Impact of social isolation and coping strategies in older adults with mild cognitive impairment during the COVID-19 pandemic: A qualitative study. Aging & Mental Health, 26(7), 1395-1416. [DOI:10.1080 /13607863.2021.1958145] [PMID]
- Farshchi, N., Kiani, Q., & Chiti, H. (2018). Effectiveness of group therapy reality in reducing depression, anxiety and increased compliance to treatment in patients with diabetic type 1. *Jour*nal of Advances in Medical and Biomedical Research, 26(117), 74-85. [Link]
- Fields, L. C., Brown, C., Skelton, J. A., Cain, K. S., & Cohen, G. M. (2021). Internalized weight bias, teasing, and self-esteem in children with overweight or obesity. *Childhood Obesity*, 17(1), 43-50. [DOI:10.1089/chi.2020.0150] [PMID] [PMCID]
- Fogelholm, M., Malmberg, J., Suni, J., Santtila, M., Kyröläinen, H., & bMäntysaari, M., et al. (2006). International physical activity questionnaire: Validity against fitness. *Medicine and Science in Sports and Exercise*, 38(4), 753-760. [DOI:10.1249/01. mss.0000194075.16960.20] [PMID]
- French, S. A., Story, M., & Perry, C. L. (1995). Self-esteem and obesity in children and adolescents: A literature review. *Obesity Research*, 3(5), 479-490. [DOI:10.1002/j.1550-8528.1995. tb00179.x] [PMID]
- Gaudiino, M., & Di Stefano, G. (2021). To detach or not to detach? The role of psychological detachment on the relationship between heavy work investment and well-being: A latent profile analysis. Current Psychology, 1-15. [DOI:10.1007/s12144-021-02259-5]
- Gordon-Larsen, P. (2001). Obesity-related knowledge, attitudes, and behaviors in obese and non-obese urban Philadelphia female adolescents. Obesity Research, 9(2), 112-118. [DOI:10.1038/oby.2001.14] [PMID]
- Guinn, B., Semper, T., Jorgensen, L., & Skaggs, S. (1997). Body image perception in female Mexican-American adolescents. The Journal of School Health, 67(3), 112-115. [DOI:10.1111/j.1746-1561.1997.tb03425.x] [PMID]
- Hajek, A., & König, H. H. (2019). The role of optimism, self-esteem, and self-efficacy in moderating the relation between health comparisons and subjective well-being: Results of a nationally representative longitudinal study among older adults. *British Journal of Health Psychology*, 24(3), 547-570. [DOI:10.1111/bjhp.12367] [PMID]

- Hajian-Tilaki, K. O., & Heidari, B. (2007). Prevalence of obesity, central obesity and the associated factors in urban population aged 20-70 years, in the north of Iran: A population-based study and regression approach. *Obesity Reviews*, 8(1), 3-10. [DOI:10.1111/j.1467-789X.2006.00235.x] [PMID]
- Halpern, B., Louzada, M. L. D. C., Aschner, P., Gerchman, F.,
  Brajkovich, I., & Faria-Neto, J. R., et al. (2021). Obesity and
  COVID-19 in Latin America: A tragedy of two pandemicsOfficial document of the Latin American Federation of Obesity Societies. Obesity Reviews, 22(3), e13165. [DOI:10.1111/obr.13165] [PMID] [PMCID]
- Hast, M. H., & Garrison, D. H. (2000). Vesalius on the variability of the human skull: Book I chapter V of De humani corporis fabrica. *Clinical Anatomy*, 13(5), 311-320. [DOI:10.1002/1098-2353(2000)13:53.0.CO;2-X] [PMID]
- Ito, T., Sugiura, H., Ito, Y., Noritake, K., & Ochi, N. (2021). Relationship between the skeletal muscle mass index and physical activity of Japanese children: A cross-sectional, observational study. PloS One, 16(5), e0251025. [DOI:10.1371/journal.pone.0251025] [PMID] [PMCID]
- Izquierdo, C. (2005). When "health" is not enough: Societal, individual and biomedical assessments of well-being among the Matsigenka of the Peruvian Amazon. Social Science & Medicine, 61(4), 767-783. [DOI:10.1016/j.socscimed.2004.08.045] [PMID]
- Jackson-Koku, G. (2016). Beck depression inventory. Occupational Medicine, 66(2), 174-175. [DOI:10.1093/occmed/kqv087] [PMID]
- Jestratijevic, I., Rudd, N. A., & Ilic, S. (2022). A body to die for: Body measurements and BMI values among female and male runway models. Clothing and Textiles Research Journal, 40(4), 291-307. [DOI:10.1177/0887302X20968893]
- Kim, M. T., Radhakrishnan, K., Heitkemper, E. M., Choi, E., & Burgermaster, M. (2021). Psychosocial phenotyping as a personalization strategy for chronic disease self-management interventions. *American Journal of Translational Research*, 13(3), 1617-1635. [PMID]
- Lemaitre, M., Chevalier, B., Jannin, A., Bourry, J., Espiard, S., & Vantyghem, M. C. (2021). Multiple symmetric and multiple familial lipomatosis. *Presse Medicale (Paris, France: 1983)*, 50(3), 104077. [DOI:10.1016/j.lpm.2021.104077] [PMID]
- Lewis, L. O. (2005). Relationship of physical activity, self-esteem, and percent body-fat to grade point average in higher education students. Tennessee: Tennessee state university. [Link]
- López-Aragón, L., López-Liria, R., Callejón-Ferre, Á. J., & Gómez-Galán, M. (2017). Applications of the standardized Nordic questionnaire: *A review. Sustainability*, 9(9), 1514. [DOI:10.3390/su9091514]
- Marmot, M. (2003). Self esteem and health. *BMJ (Clinical Research ed.)*, 327(7415), 574-575. [DOI:10.1136/bmj.327.7415.574] [PMID] [PMCID]
- Mazereel, V., Vansteelandt, K., Menne-Lothmann, C., Decoster, J., Derom, C., & Thiery, E., et al. (2021). The complex and dynamic interplay between self-esteem, belongingness and physical activity in daily life: An experience sampling study in adolescence and young adulthood. *Mental Health and Physical Activity*, 21, 100413. [DOI:10.1016/j.mhpa.2021.100413]

- Mazza, M. G., De Lorenzo, R., Conte, C., Poletti, S., Vai, B., & Bollettini, I., et al. (2020). Anxiety and depression in COV-ID-19 survivors: Role of inflammatory and clinical predictors. *Brain, Behavior, and Immunity, 89,* 594-600. [DOI:10.1016/j.bbi.2020.07.037] [PMID] [PMCID]
- McCarthy, S. N., Gibney, M. J., Flynn, A., & Irish Universities Nutrition Alliance. (2002). Overweight, obesity and physical activity levels in Irish adults: evidence from the North/South Ireland food consumption survey. *The Proceedings of the Nu*trition Society, 61(1), 3–7. [DOI:10.1079/PNS2001121] [PMID]
- McCoy, S. M., & Morgan, K. (2020). Obesity, physical activity, and sedentary behaviors in adolescents with autism spectrum disorder compared with typically developing peers. *Autism*, 24(2), 387-399. [DOI:10.1177/1362361319861579] [PMID]
- Moghaddam, M. B., Aghdam, F. B., Jafarabadi, M. A., Allahver-dipour, H., Nikookheslat, S. D., & Safarpour, S. (2012). The Iranian Version of International Physical Activity Questionnaire (IPAQ) in Iran: Content and construct validity, factor structure, internal consistency and stability. World Applied Sciences Journal, 18(8), 1073-1080. [Link]
- Montero-Simó, M. J., Araque-Padilla, R. A., Melero-Bolaños, R., & Shultz, II, C. J. (2022). The role of Spain's Catalytic Institutions to facilitate community well-being during the COVID-19 pandemic. In C.J. Shultz, II, D.R. Rahtz, & M. J. Sirgy (Eds), Community, economy and COVID-19 (pp. 461-488). Cham: Springer. [DOI:10.1007/978-3-030-98152-5\_22]
- Morrill, C. M., Leach, J. N., Shreeve, W. C., Radebaugh, M. R., & Linder, K. (1991). Teenage obesity: An academic issue. *International Journal of Adolescence and Youth*, 2(4), 245-250. [DOI:10.1080/02673843.1991.9747683]
- Morrison, T. L., Thomas, M. D., & Weaver, S. J. (1973). Self-esteem and self-estimates of academic performance. *Journal of Consulting and Clinical Psychology*, 41(3), 412-415. [DOI:10.1037/h0035337] [PMID]
- Namnik, N., Negahban, H., Salehi, R., Shafizadeh, R., & Tabib, M. S. (2016). Validity and reliability of Persian version of the Specific Nordic questionnaire in Iranian industrial workers. *Work*, *54*(1), 35-41. [DOI:10.3233/WOR-162268] [PMID]
- Narimani, M., & Mousazadeh, T. (2010). Comparing self-esteem and self-concept of handicapped and normal students. *Procedia-Social and Behavioral Sciences*, 2(2), 1554-1557. [DOI:10.1016/j.sbspro.2010.03.234]
- Nayak, J., Mishra, M., Naik, B., Swapnarekha, H., Cengiz, K., & Shanmuganathan, V. (2022). An impact study of COVID-19 on six different industries: Automobile, energy and power, agriculture, education, travel and tourism and consumer electronics. *Expert Systems*, 39(3), e12677. [DOI:10.1111/exsy.12677] [PMID]
- Onsori, M., & Galedari, M. (2015). Effects of 12 weeks aerobic exercise on plasma level of TSH and thyroid hormones in sedentary women. *European Journal of Sports and Exercise Science*, 4(1), 45-49. [Link]
- Páez-Maldonado, J. A., Reigal, R. E., Morillo-Baro, J. P., Carras-co-Beltrán, H., Hernández-Mendo, A., & Morales-Sánchez, V. (2020). Physical fitness, selective attention and academic performance in a pre-adolescent sample. *International Journal of Environmental Research and Public Health*, 17(17), 6216. [PMID] [PMCID]

- Pascoe, M. C., J. de Manincor, M., Hallgren, M., Baldwin, P. A., Tseberja, J., & Parker, A. G. (2021). Psychobiological mechanisms underlying the mental health benefits of yoga-based interventions: A narrative review. *Mindfulness*, 12(12), 2877-2889. [DOI:10.1007/s12671-021-01736-z]
- Pesa, J. A., Syre, T. R., & Jones, E. (2000). Psychosocial differences associated with body weight among female adolescents: The importance of body image. *The Journal of Adolescent Health: Official Publication of the Society for Adolescent Medicine*, 26(5), 330–337. [DOI:10.1016/S1054-139X(99)00118-4] [PMID]
- Ralapanawa, U., & Sivakanesan, R. (2021). Epidemiology and the magnitude of coronary artery disease and acute coronary syndrome: A narrative review. *Journal of Epidemiology and Global Health*, 11(2), 169-177. [DOI:10.2991/jegh.k.201217.001] [PMID] [PMCID]
- Ramos de Marins, V. M., Varnier Almeida, R. M., Pereira, R. A., & Barros, M. B. (2001). Factors associated with overweight and central body fat in the city of Rio de Janeiro: Results of a two-stage random sampling survey. *Public Health*, 115(3), 236-242. [DOI:10.1038/sj.ph.1900763] [PMID]
- Reddy, S. P., Mbewu, A. D., Williams, D. R., Harriman, N. W., Sewpaul, R., & Morgan, J. W., et al. (2021). Race, geographical location and other risk factors for hypertension: South African National Health and Nutrition Examination Survey 2011/12. SSM-Population Health, 16, 100986. [PMID] [PMCID]
- Rosenberger, M. E., Fulton, J. E., Buman, M. P., Troiano, R. P., Grandner, M. A., & Buchner, D. M., et al. (2019). The 24-hour activity cycle: A new paradigm for physical activity. *Medicine and Science in Sports and Exercise*, 51(3), 454–464. [DOI:10.1249/MSS.0000000000001811] [PMID] [PMCID]
- Ryan, M. P. (2008). The antidepressant effects of physical activity: Mediating self-esteem and self-efficacy mechanisms. *Psychology & Health*, 23(3), 279-307. [PMID]
- Sánchez-SanSegundo, M., Alarcó-Rosales, R., Zaragoza-Martí, A., Quesada-Rico, J. A., Gabaldón-Bravo, E., & Hurtado-Sánchez, J. A. (2022). The associations of mental health disturbances, self-esteem, and partner violence victimization with condom use in Spanish adolescents. *Journal of Clinical Medicine*, 11(9), 2467. [DOI:10.3390/jcm11092467] [PMID] [PMCID]
- Scherrer, V., & Preckel, F. (2019). Development of motivational variables and self-esteem during the school career: A meta-analysis of longitudinal studies. *Review of Educational Research*, 89(2), 211-258. [DOI:10.3102/0034654318819127]
- Sinha, A., & Kumari, S. (2022). Effect of short duration integrated classroom yoga module on physical fitness, cognitive performance, emotional well being and personality characteristic measures of school children. Voga Mimamsa, 53, 100-108. [DOI:10.4103/ym.ym\_55\_21]
- Soltani, M., Rashidlamir, A., Fathei, M., & Ghahremanimoghaddam, M. (2018). [The effect of eight weeks of water training on Sirt1, Pgc-1α and body fat percentage in obese men (Persian)]. Journal of Babol University of Medical Sciences, 20(9), 55-60. [Link]
- Speelman, A. D., Van De Warrenburg, B. P., Van Nimwegen, M., Petzinger, G. M., Munneke, M., & Bloem, B. R. (2011). How might physical activity benefit patients with Parkinson disease? *Nature Reviews Neurology*, 7(9), 528-534. [DOI:10.1038/ nrneurol.2011.107] [PMID]

- Su, J., Li, Q., Mao, P., Peng, H., Han, H., & Wiley, J., et al. (2022). Does the association of sedentary time or fruit/vegetable intake with central obesity depend on menopausal status among women? *International Journal of Environmental Research* and Public Health, 19(16), 10083. [PMID] [PMCID]
- Suss, A. L., Tinkelman, B. K., Freeman, K., & Friedman, S. B. (1996). School attendance, health-risk behaviors, and selfesteem in adolescents applying for working papers. *Bulletin* of the New York Academy of Medicine, 73(2), 255-266. [PMID]
- Tan, S. T., Tan, C. X., & Tan, S. S. (2021). Physical activity, sedentary behavior, and weight status of university students during the COVID-19 lockdown: A cross-national comparative study. *International Journal of Environmental Research and Public Health*, 18(13), 7125. [DOI:10.3390/ijerph18137125] [PMID] [PMCID]
- Tandon, P. S., Zhou, C., Johnson, A. M., Gonzalez, E. S., & Kroshus, E. (2021). Association of children's physical activity and screen time with mental health during the COVID-19 pandemic. *JAMA Network Open*, 4(10), e2127892. [PMID] [PMCID]
- Trockel, M. T., Barnes, M. D., & Egget, D. L. (2000). Health-related variables and academic performance among first-year college students: Implications for sleep and other behaviors. *Journal of American College Health*, 49(3), 125-131. [DOI:10.1080/07448480009596294] [PMID]
- Wang, F., Wild, T. C., Kipp, W., Kuhle, S., & Veugelers, P. J. (2009). The influence of childhood obesity on the development of self-esteem. *Health Reports*, 20(2), 21–27. [PMID]
- Wang, Y., Zhao, L., Gao, L., Pan, A., & Xue, H. (2021). Health policy and public health implications of obesity in China. *The* lancet Diabetes & Endocrinology, 9(7), 446-461. [DOI:10.1016/ S2213-8587(21)00118-2] [PMID]
- Wijngaarde, C. A., Veldhoen, E. S., van Eijk, R. P. A., Stam, M., Otto, L. A. M., & Asselman, F. L., et al. (2020). Natural history of lung function in spinal muscular atrophy. *Orphanet Journal of Rare Diseases*, 15(1), 88. [PMID] [PMCID]
- Wilsgaard, T., Jacobsen, B. K., & Arnesen, E. (2005). Determining lifestyle correlates of body mass index using multilevel analyses: the Tromsø Study, 1979-2001. *American Journal of Epidemiology*, 162(12), 1179-1188. [DOI:10.1093/aje/kwi328] [PMID]
- Yang, F., Sun, J., Li, J., & Lyu, S. (2022). Coping strategies, stigmatizing attitude, and cyberbullying among Chinese college students during the COVID-19 lockdown. Current Psychology (New Brunswick, N.J.), 1–9. Advance Online Publication. [PMID]
- Yuan, Y., Ji, X., Yang, X., Wang, C., Samsudin, S., & Omar Dev, R. D. (2022). The effect of persistence of physical exercise on the positive psychological emotions of primary school students under the STEAM education concept. *International Jour*nal of Environmental Research and Public Health, 19(18), 11451. [PMID] [PMCID]

